Author's response to reviews

Title: Institutional delivery service utilization in Munisa Woreda, South East Ethiopia: A community based cross-sectional study.

Authors:

Abdella A Abdo (a2ay.or@gmail.com)
Abebaw G Worku (gabebaw2worku@gmail.com)
Zelalem B Mengesha (zelalem78@gmail.com)

Version: 5 Date: 29 July 2012

Author's response to reviews: see over
To- Chief Editor of BMC- Pregnancy and childbirth

Dear Editor;

We were requested to do revisions on English usage and grammar as there were a number of errors remaining. Accordingly, we gave the manuscript for language expert (Language editor of the Ethiopian Journal of Health and Biomedical sciences) to be copy edited. The final language edited version and another document with the track changes are attached.

With best regards

Abdella Amano

Corresponding Author